Author’s response to reviews

Title: Associations between intimate partner violence, childcare practices and infant health: Findings from Demographic and Health Surveys in Bolivia, Colombia and Peru

Authors:

Helga B Urke (helga.urke@uib.no)
Maurice B Mittelmark (maurice.mittelmark@uib.no)

Version: 4 Date: 10 August 2015

Author’s response to reviews:

Dear editor,

We are grateful for another review of the paper and have taken care in following the recommendations of the reviewer. Please see below for detailed responses to the reviewer’s comments.

Reviewer:
Although authors remarkably improved the manuscript by modifying it based on my last comments, still I have found several mistakes. Particularly, they mismatched the texts and tables. Although they presented results of logistic regression analyses in Table 5, they mentioned that these results are presented in table 2. It is surprising that authors were not careful during revision.

Authors’ response: We are embarrassed about the mistakes of mismatch between table number and text. We have now taken great care to correct it.

Reviewer:
Somehow, the text based on bivariable analysis is too long (almost three pages), which should be reduced.

Authors’ response: We agree with this comment. The text based on bivariate analyses is shortened to approximately 1,5 pages.

Reviewer:
There are also some minor mistakes that should be thoroughly corrected.

Authors’ response: We have done a thorough proofread of the entire manuscript and corrected any mistakes.